Author's response to reviews

Title: Clinical and economic evaluation of Spanish 65-year-cohort pneumococcal vaccination measured by a dynamic model

Authors:

Roberto Pradas (rpradasvelasco@gmail.com)
Angel Gil de Miguel (angel.gil@urjc.es)
Alejandro Alvaro (alejandro.alvaro@urjc.es)
Ruth Gil-Prieto (ruth.gil@urjc.es)
Reyes Lorente (maria-reyes.lorente@unirioja.es)
Cristina Mendez (cristina.mendezdiez@pfizer.com)
Pablo Guijarro (pablo.guijarromasi@pfizer.com)
Fernando Antoñanzas (fernando.antonanzas@unirioja.es)

Version: 2 Date: 2 November 2012

Author's response to reviews: see over
Dear editor,

Madrid, November 2012

As email received we have made some changes on manuscript version

1. Competing Interest added to manuscript
2. Author’s Contribution added to manuscript.
3. Initial manuscript was removed and uploaded new version.

No more changes were made.

If you require any further information, please do not hesitate to contact me:

Sincerely,

Pablo Guijarro Masi

Pfizer Spain,
Avda. de Europa, 20 B
Parque Empresarial La Moraleja
28108 Alcobendas (Madrid)
Tfn: (+34) 91 490 92 98
Mobile: (+34) 600 911 618
Fax.: (+34) 91 490 97 04
E-mail. pablo.guijarromasi@pfizer.com